Author's response to reviews

Title: A complex nursing intervention on Complementary and Alternative Medicine (CAM) to increase quality of life in patients with breast and gynaecologic cancer undergoing chemotherapy: study protocol for a partially randomized patient preference trial

Authors:

Nadja Klafke (nadja.klafke@med.uni-heidelberg.de)
Cornelia Mahler (cornelia.mahler@med.uni-heidelberg.de)
Cornelia von Hagens (cornelia.von.hagens@med.uni-heidelberg.de)
Justine Rochon (rochon@imbi.uni-heidelberg.de)
Andreas Schneeweiss (andreas.schneeweiss@med.uni-heidelberg.de)
Andreas Müller (andreas.mueller.fk@klinikum-karlsruhe.de)
Hans-Joachim Salize (hans-joachim.salize@zi-mannheim.de)
Stefanie Joos (stefanie.joos@med.uni-heidelberg.de)

Version: 5
Date: 19 December 2014

Author's response to reviews:

Nadja Klafke
Department of General Practice and Health Services Research
University Hospital Heidelberg
Voßstr. 2
D-69115 Heidelberg
Phone: +49 6221 56 4818
Fax: +49 6221 56 1972
Email: nadja.klafke@med.uni-heidelberg.de

December 19, 2014

Professor Doug Altman
Editor-in-Chief
Trials
c/o BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Submission of the final version of the manuscript titled “A complex nursing
intervention on Complementary and Alternative Medicine (CAM) to increase quality of life in patients with breast and gynaecologic cancer undergoing chemotherapy: study protocol for a partially randomised patient preference trial”

Dear Professor Altman,

Dear Reviewers,

Thank you for again reviewing our revised manuscript and letting us know that you are fine with the revisions we have made.

We have formatted the manuscript according to the Journal’s standard and have given it a final spell and read check.

We are now confident that the manuscript is ready for publication. Please find attached the final documents.

Many thanks in advance.

With kindest regards and best wishes for the Christmas season.

Yours sincerely,

Nadja Klafke